**Author's response to reviews**

**Title:** Acute febrile neutrophilic dermatosis (Sweet's syndrome) in a child, associated with a Rota virus infection: a case report

**Authors:**

Alexandros Makis (amakis@cc.uoi.gr)
Stavros Stavrou (sstavrou77@gmail.com)
Nikolaos Chaliasos (nhalias@cc.uoi.gr)
Aikaterini Zioga (aikaterini.zioga@medai.gr)
Antonios P Vlahos (anvlahos@uoi.gr)
Georgios Gaitanis (ggaitan@cc.uoi.gr)
Antigone Siamopoulou (asiamop@cc.uoi.gr)
Ioannis D Bassukas (ibassuka@cc.uoi.gr)

**Version:** 3 **Date:** 29 May 2010

**Author's response to reviews:**

We would like to thank you for your helpful comments.

**Major revisions**

Patient ID: The patient identification (age, gender and ethnicity) has been stated consistently in the manuscript and the abstract.

Figures: As you may already know, we had technical problems with the submission of the figures and the Journal’s team kindly assisted me with the uploading. We would be grateful if you could help us again with the resubmission of the figures at this point.

Tables: Tables 2 and 3 have been reformatted into an appendix. Table 1 has been amended.

References: References have been written according to instructions

**Minor revisions**

Title: the title ends `: a case report`

All highlighting, tracking or coloured text has been removed from the manuscript.

Authors contributions have been formatted into a paragraph

Structure and Typography have been checked and corrected according to instructions.

Many thanks

Alexandros Makis